Author's response to reviews

Title: Triple negative breast cancers express receptors for LHRH and are potential therapeutic targets for cytotoxic LHRH-analogs AEZS 108 and AEZS 125

Authors:

Stephan Seitz (seitzs@caritasstjosef.de)
Stefan Buchholtz (sbuchholtz@caritasstjosef.de)
Andrew V Schally (avschally@va.gov)
Florian Weber (florian.weber@ukr.de)
Monika Klinkhammer-Schalke (zentrum.tumor@klinik.uni-regensburg.de)
Elisabeth C Inwald (einwald@caritasstjosef.de)
Roberto Perez (robert.perez.phd@gmail.com)
Ferenc G Rick (Frick@med.miami.edu)
Luca Szalontay (Lszalontay@miami.edu)
Florian Hohla (f.hohla@salk.at)
Sabine Segerer (ssegerer@endokrinologikum.de)
Chiu-Wai Kwok (babsckw@hotmail.com)
Olaf Ortmann (ortmann@caritasstjosef.de)
Jörg B Engel (joergbengel@hotmail.com)

Version: 4 Date: 17 August 2014

Author's response to reviews:

MS: 7965861941136797
Triple negative breast cancers express receptors for LHRH and are potential therapeutic targets for cytotoxic LHRH-analogs AEZS 108 and AEZS 125

Editors comments:
1. Please could you move your references to just before your tables and figure legends.
Done
2. Please could you include the following sub headings in your abstract:- Background, Methods, Results, Conclusions.
Done
3. Please rename your Introduction' section as 'Background'.
Done
4. Please can you clarify whether an animal ethics committee approved your study and provide the full name of this committee in your manuscript.
Animal experiments were approved by the local ethics committee of Veterans Administration Hospital, Miami, FL.
5. In addition, please include the name of the ethics committee which approved your use of tumour samples for the purposes of your study.
Experiments was approved by the local ethics committee of the Medical University of Regensburg.

6. Please clarify whether written informed consent was obtained from patients for the use of the tumour samples.
Written informed consent was obtained by all patients.

7. Please note that some words are merged together in your manuscript, please check spacing of words throughout your whole manuscript and amend accordingly.
Done

Dear editors,
All points mentioned by you were addressed point to point. We hope that the MS is now adequate for publication in your prestigious journal. We thank you and the reviewers for their constructive remarks.

Yours sincerely
Jörg B. Engel, MD